Author's response to reviews

Title: Out of hours care: A profile analysis of patients attending the emergency department and the general practitioner on call.

Authors:

Hilde Philips (hilde.philips@ua.ac.be)
Roy Remmen (roy.remmen@ua.ac.be)
Peter De Paepe (peter.depaepe@ugent.be)
Walter Buylaert (walter.buylaert@ugent.be)
Paul Van Royen (paul.vanroyen@ua.ac.be)

Version: 4 Date: 3 November 2010

Author's response to reviews:

Dear editor,

Please find the revision of our manuscript hereby attached. We made all adjustments as requested.

We sent additional files together with an earlier version of the manuscript, to clarify some questions of one of the referees. Because they are not in English (questionnaires as we used them during our research), they do not have any additional value to the reader of the manuscript. Therefore, they are not added to the manuscript.

Kind regards,

Hilde Philips